Author’s response to reviews

Title: Extensive haemorrhagic necrosis of liver is an unpredictable fatal complication in dengue infection: a postmortem study

Authors:

S.A.M. Kularatne (samkul@slt.net.lk)
I.V.B. Imbulpitiya (dhin_im@yahoo.com)
R.A. Abeysekera (rajithaasa55@gmail.com)
R. Waduge (roshitha2006@yahoo.com)
R.P.V.J. Rajapakse (jayanthar@pdn.ac.lk)
K.G.A.D. Weerakoon (kosalagadw83@gmail.com)

Version: 2
Date: 27 November 2013

Author’s response to reviews:

The editors,

BMC Infectious Diseases

We wish to state that there has been no funding of any type given to this research by any organization.

As revision, we have included a title page with author details at the beginning of the manuscript.

Secondly the 2nd author name and email address is corrected as follows:

Name: I.V.B. Imbulpitiya
email: dhin_im@yahoo.com

Thank you.

Yours trustfully,

Dr. R.A. Abeysekera

Corresponding author